



A Review of Advanced Control Strategies of Microgrids with Charging Stations

Matej Tkac, Martina Kajanova *D and Peter Bracinik D

Faculty of Electrical Engineering and Information Technology, University of Zilina, 010 26 Zilina, Slovakia

* Correspondence: martina.kajanova@uniza.sk

Abstract: In the context of the global drive towards sustainability and rapid integration of renewables, electric vehicles, and charging infrastructure, the need arises for advanced operational strategies that support the grid while managing the intermittent nature of these resources. Microgrids emerge as a solution, operating independently or alongside the main grid to facilitate power flow management among interconnected sources and different loads locally. This review paper aims to offer a comprehensive overview of the different control strategies proposed in the literature to control microgrids with electric vehicle charging stations. The surveyed research is primarily categorized according to the employed control algorithms, although distinctions are also made based on defined microgrid architecture, utilization of specific power sources, and charging stations configurations. Additionally, this paper identifies research gaps in the current research. These gaps encompass the use of oversimplified models for charging stations and/or renewable sources operation, limited simulation time periods, or lack of experimental testing of proposed approaches. In the light of these identified shortcomings, this manuscript presents recommendations for guiding future research.

Keywords: microgrid control; charging station; renewable energy resources; control strategies

1. Introduction

Top topics of our lives nowadays include decarbonization and environmentally friendly lifestyles. This movement towards sustainability also impacts the operation of electrical power systems, presenting various challenges for their efficient management and control. As the world strives to integrate a growing proportion of renewable energy sources and reduce reliance on fossil fuels, traditional power grids encounter stability issues due to the fluctuating nature of renewables. Moreover, changes in energy consumption patterns among consumers introduce added complexity to the equation.

To ensure a seamless and sustainable integration of renewable energy sources into the power grid, addressing these challenges is of paramount importance. Innovative approaches, such as demand-side management and advanced grid control algorithms, hold the key to adapting power systems to the dynamic nature of renewable energy generation and consumer demand. In this regard, the concept of the smart grid has emerged, aiming to create a reliable, efficient, and sustainable energy infrastructure by leveraging modern communication, automation, and information technologies [1].

Microgrids, as a specialized form of smart grid, have gained popularity in recent years. These autonomous and localized energy systems operate independently or in coordination with the main grid to generate, distribute, and manage electricity for specific geographic areas of interconnected consumers [2]. With their ability to function as selfcontained entities, equipped with their own energy resources, control mechanisms, and often energy storage capabilities, microgrids offer promising solutions for energy resilience and sustainability.

Managing microgrids effectively is critical to realizing their full potential. By ensuring efficient coordination of energy resources and controlling energy flow, microgrids can



Citation: Tkac, M.; Kajanova, M.; Bracinik, P. A Review of Advanced Control Strategies of Microgrids with Charging Stations. *Energies* **2023**, *16*, 6692. https://doi.org/10.3390/ en16186692

Academic Editor: GM Shafiullah

Received: 7 August 2023 Revised: 12 September 2023 Accepted: 15 September 2023 Published: 18 September 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). effectively balance supply and demand, minimize energy waste, and optimize energy consumption patterns. Furthermore, advanced control strategies and tools are required to guarantee the reliability and resiliency of microgrids, especially in light of their growing association with the electric vehicle (EV) revolution.

The rising popularity of electric vehicles as a means to reduce carbon emissions in the transportation sector introduces a new dimension of complexity to the energy landscape [3]. The increasing number of EVs and the time windows of their charging sessions significantly impact the operation of power systems, necessitating more sophisticated control strategies to align the charging patterns of EVs with grid conditions. This close interplay between electrical power systems and the transportation sector underscores the need for advanced control measures to ensure the seamless integration of EVs into microgrids, safeguarding stable and resilient power supply to consumers.

Numerous literature reviews address the charging of electric vehicles. In Savio Abraham's work [4], the authors present connector types, architectural configurations of charging stations, and control algorithms proposed for charging control. However, the role of the charging station in the electric power systems or microgrids is usually absent. There are also reviews for specific control algorithms. Nimalsiri [5] undertakes a comprehensive evaluation of decentralized control algorithms, while [6] focuses on hierarchical control algorithms. However, both [5,6] neglect the critical influence of the charging station on grid operation—be it conventional or microgrid.

There are also reviews that deal with the role of EV charging stations within the electrical grid. The review in [7] focuses on EVs and their charging stations in energy and transportation systems. That review also addresses market-driven charging station management. However, [7] fails to address the effects of charging stations on microgrid stability and operation. Similarly, the study in [8] deals with the optimal management strategies in grids with electric vehicles, but the specifics of microgrids with EV integrations are omitted.

There are also reviews that focus on the microgrid's operation but most of these studies neglect the effects of EV charging stations. Vadi's work [9], for instance, critically evaluates control optimization methods for transient stability in microgrids, yet this work sidesteps exploration of how EV charging stations affect microgrid operations. Likewise, [10] dissects reinforcement learning methods in the power management of grid-tied microgrids as a tool more adaptive to stochasticity in such grids. However, there is no information about the effect of EV charging stations on microgrid operation or on the islanded microgrids' control algorithms. On the other hand, the review in [11] studies the role of EVs in the microgrid. The authors in [11] deal with design, control, energy management, and protection in DC microgrids for EV charging. A review of AC or hybrid microgrids is missing, and even more so, the primary focus of these microgrids is to supply required power to the charging station and coverage of other loads is neglected. There is also a review in [12] that studies the role of the so-called fleet operator that can control charging groups of EVs. However, independent usage of personal EVs is neglected.

This review aims to discuss the role of EV charging stations in microgrid operation more generally. This paper presents studies with different microgrid architectures and various control algorithms, as well as the different roles of EV charging stations in microgrid operational strategies. The main contribution of this paper is the definition of the current research shortcomings and the research gaps that should be addressed in future research to enable the implementation of such advanced operational strategies in real-world applications.

The rest of this paper is organized as follows: Section 2 presents the review methodology and statistics of the presented research studies. Next, Section 3 presents the research studies that focus on power control in the microgrids with charging stations. Similarly, Section 4 discusses the studies that deal with voltage control and Section 5 focuses on the research papers covering the economic operation control of microgrids. The studies with other control strategies than the previous ones are presented in Section 6. Section 7 discusses the research gaps in presented manuscripts and defines recommendations for future work. Section 8 concludes the paper.

2. Review Methodology and Statistics

The compilation of scientific papers for this review article was based on searching for certain keywords within scientific databases. The keywords applied for the search in scientific databases were "microgrid", "control", "charging", and "station", thereby ensuring the inclusion of articles that incorporated any of these terms. The scientific databases used for the search included Web of Science, Scopus, and IEEE Xplore with the selected articles discussing the topic of microgrid control with charging stations for electric vehicles. During the selection of relevant literature, we followed a similar process to the one presented in [13], with the literature selection going through several steps. In the initial step, the aforementioned keywords were utilized for search in the scientific databases. The second step involved reading the titles and abstracts of the articles obtained from the search in the scientific databases. This process led to a compilation of a set of literature for each scientific database. Subsequently, the individual sets of literature were merged into a single set from which duplicate papers were removed. The final step in the literature selection process entailed the full-text reading of the articles, based on which the final set of literature utilized for this review article was compiled. The final set of publications presents, from the author's perspective, the most relevant articles to the topic discussed in this review article. To help with the discussion and division of reviewed literature, a number of statistics were made and are shown in this section.

2.1. Reviewed Publications

The final set of publications consists of 70 papers which were published in journals as well as papers from conferences. The publications selected for this review span from the year 2010 to the present day. In recent years, there has been an increase in the frequency of publications related to microgrid control with charging stations for electric vehicles. This recent increase in interest in this topic may be caused by the increase in the number of EVs as well as in the number of corresponding charging stations, or as a response to many initiatives planned as a tool to reduce greenhouse gas emissions. The frequency of selected publications through the years is shown in Figure 1. The graph representing the distribution of journal papers and conference papers among the final set of papers is shown in Figure 2 where it can be seen that the majority of papers discussed in this review are from journals. Figure 3 shows the distribution of different publishers among the selected set of papers, as well as between journals and conference papers. From this graph, it is apparent that most of the journal papers in our selection are from ELSEVIER and MDPI, while almost all conference papers are from IEEE. It is important to note that the paper selection was conducted at the end of the second quarter of 2023.



Figure 1. Publications by years.

Journal Conference

Figure 2. Type of publication.





2.2. Types and Architectures of Microgrids

Research in the utilization of microgrids has in recent years gained more interest due to the development and utilization of renewable energy sources (RESs), as well as electric vehicle charging stations. Utilization of microgrids provides a flexible solution for the implementation of these facilities and at the same time provides space for the implementation of novel control structures. Microgrids are typically divided into three types based on the characteristic of currents and voltages used by the loads in the microgrid. These types are AC microgrid, DC microgrid, and Hybrid microgrid. The architecture of a microgrid is based on the interconnection between the distribution network and the microgrid. The most common architectures in literature are grid-connected microgrids and islanded microgrids. The distribution of the different types and architectures of microgrids in our final set of papers is shown in Figure 4.

In the case of microgrid types (Figure 4a), the majority of microgrids were operated as DC microgrids, and in the case of microgrid architectures (Figure 4b), the majority of microgrids were operated in grid-connected mode. In a few of the papers [14,15], the authors tested their control mechanisms in grid-connected as well as in islanded mode to prove the reliability of their systems. During the full read of some papers, it was not clearly stated or defined what type of microgrid or architecture of microgrid was used in these papers. Therefore, in these cases, we used the abbreviation of the phrase "not defined (n.d.)" to visualize these papers in graphs.



Figure 4. Distribution of types (a) and architectures (b) of microgrids.

2.3. Energy Generation and Storage in Microgrids

One of the advantages of microgrids is the possibility for implementation of complementary power sources to enable the independent operation of the microgrid system from the utility grid. This possibility has enabled increased research in the area of microgrid implementation. Complementary power sources used in microgrid systems are either renewable energy sources such as photovoltaic systems (PV), wind turbines (WT), fuel cells (FC), and small hydro generators (SHG), or conventional energy sources powered by fossil fuels such as diesel generators (DG), microturbines (MT), and combined heat and power (CHP). The utilization of renewable energy sources is the most common method utilized for independent energy generation where solar and wind power are most commonly used. RESs also provide a possibility to reduce greenhouse gas emissions associated with energy production. The distribution of different energy sources used in the final set of review papers can be seen in Figure 5. From the final set of papers, only three papers did not utilize RESs, as it is possible to see from Figure 5a representing the distribution of individual RESs among the final set of papers. On the other hand, conventional energy sources (CESs) were not utilized in 55 papers from the final set. The rest of the papers either had one or more RESs or CESs. In the case of one paper, the authors have not stated clearly the type of RES, so therefore to represent this case in a graph we used the same abbreviation "n.d." as in the previous section. To represent a situation when no RESs or CESs were used in a discussed paper, the abbreviation of the phrase "not present (n.p.)" is used to visualize these papers in graphs.



Figure 5. Distribution of renewable energy sources (a) and conventional energy sources (b).

Another important part of microgrids in combination with RESs is the use of energy storage systems. Energy storage systems (ESS) are used primarily for the storage of excess

power produced by the RESs and the subsequent use of this energy to support microgrids themselves, or individual loads placed in microgrids, or to provide support to the utility grid. In addition to the power flow exchange, the storage system can be used for power quality or microgrid stability control. The most common storage system is the battery storage system (BSS) which was used in 48 papers as shown in Figure 6. Other storage systems used in the literature are hydrogen storage systems (HSS), supercapacitors (SC), and fuel cell energy storage (FCES). Similarly, for complementary power sources, not all review papers have energy storage systems implemented in their respective microgrids. Therefore, to represent these papers in graphs, we used the abbreviation "n.p." as defined in a previous section.



■ Battery storage ■ Hydrogen storage ■ Supercapacitor ■ FCES ■ n.p.

Figure 6. Distribution of different energy storage systems.

2.4. Loads and Charging Stations in Microgrids

There has been a significant increase in the adoption of electric vehicles in recent years as a result of international efforts to reduce greenhouse gas emissions and promoting sustainable transportation solutions. An increase in the numbers of electric vehicles consequently leads to an increase in the numbers of charging stations [16]. To prevent problems in electrical networks caused by the irregular nature of charging station consumption (electric vehicles charging) or to exploit advantages of energy transfer between electric vehicles and charging stations (vehicle to grid (V2G) concept), installations of electric vehicle charging stations within microgrids are being used and researched. In reviewed papers, different charging stations were mentioned and used in the control structure of microgrids. The distribution of different types of charging stations is shown in Figure 7b. The most used charging station type in reviewed papers was the AC charging station (CS), followed by the fast charging station (FCS), and the third most used type was slow DC charging. In 12 reviewed papers, the type of charging station was not clearly defined, and therefore to represent these cases we again use the abbreviation "n.d.". Additionally, charging stations with other loads (Figure 7a) were also implemented into the microgrids in a total of 40 papers from the final set. These loads represent residential loads in the form of houses or apartment buildings, workplace loads, or industrial loads that were connected to the discussed microgrids.



Figure 7. Distribution of loads (a) and different charging station types (b) in microgrids.

2.5. Control Strategies in Microgrids

The utilization of a microgrid system allows for implementing various control strategies that can improve microgrid operation, increase the utilization RESs, and maximize the economic benefit or other aspects associated with a microgrid system utilization. During the literature review, we have identified three major control objectives most frequently occurring in the literature. These objectives are power control, voltage control, and economic operation control in addition to other less frequently occurring control objectives (frequency control, current control, etc.). Figure 8 shows the distribution of different control objectives amongst the final set of papers discussed in this review. It is important to note that in some papers multiple control objectives were discussed and implemented to increase operational control or to compare multiple control strategies. A closer discussion of each major control objective is presented in the following sections.



Figure 8. Distribution of control strategies in microgrids.

3. Power Control Strategies of Microgrids

Implementation of power control brings a number of possibilities which can improve microgrid operation. Power control in microgrid operation can improve the utilization of energy sources within the microgrid, minimize the dependence on the utility grid, or even manage the power flow. Control objectives used in combination with power control are discussed in the following subsections.

3.1. Power Flow Control of Microgrids

Power flow control strategy represents a method used for the optimal distribution of power between energy sources connected to the microgrid, utility grid, ESS, loads, and EVs connected to the microgrid through CS. Optimal distribution of power can improve the power balance, power quality, or stability of a microgrid, and in a microgrid with electric

vehicle charging stations (EVCS) it can ensure that the EVs connected to the CS are charged to an optimal level. To achieve this, researchers are using different control methods or control algorithms, as well as different configurations of microgrid systems.

Most of the reviews studies with power flow control are focused on DC microgrids. In Sayed's work [17], the authors focused on energy management for a DC microgrid system in islanded mode to control power distribution from different RESs to EVs connected to the CS within a microgrid. Charging of connected EVs is regulated based on the amount of power produced from RES and energy stored in battery storage. Power flow management with a fuzzy logic controller in [18] is used for voltage stabilization in a DC microgrid for EVCS. The proposed controller is used to coordinate power flow between the PV power plant, BSS, and the utility grid based on variations in EVCS demand and intermittent power generation. The fuzzy logic controller was also used in [19] for a supervisory control system to achieve power balance between the energy supply and EVCS within a DC microgrid. In García-Triviño's work [20], a fuzzy logic controller is used for a decentralized energy management system to control power flow, bus voltage, and battery storage state of charge. The DC microgrid proposed in this paper consisted of FCS, PV system, and battery energy storage with connection to the utility grid. The control proposed in [20] was used to control the power flow of a PV system, BSS, and utility grid to supply the desired amount of power for the charging of EVs connected to FCS. Whit proposed how control of the FCS was able to work in a stand-alone mode for extended time periods. In Abraham's works [21,22], a strategy for the power management of reliable power supply was introduced. The proposed power management is based on fuzzy logic control, which directs bidirectional power flow between EVs, BSS, and utility grid. Vehicle-to-vehicle (V2V) concept was used in these papers to provide share power between EVs when necessary. In Naik's work [23], a smart energy management strategy is proposed to tackle issues of intermittent nature of PV power generation and EV load as well as ensure sustainable power flow control for EV charging stations connected to DC microgrids. Presented energy management used the control of BSS and small hydro generators to meet the peak EV load demand as well as to reduced dependency on the utility grid. Energy management system in [24] is presented for a hybrid system of DC microgrid with multi-battery storage system, a grid-tie inverter with a PV installation, and FCS. The energy management system in this paper is responsible for energy exchange control with the utility grid as well as for allocating the battery strings to different components of the hybrid system. String allocation control must ensure that the battery connected to this string can successfully meet the EV charging demand. Paper [25] presents an energy management strategy for the supply infrastructure of EVCS placed in a DC microgrid. The control system aims to control energy exchange between microgrid and utility grid and ensure reliable service to the users from cost-effective energy supply to EVs. The control system also controls charging/discharging of EVs to reduce power purchases from the utility grid or to take advantage of selling prices to the utility grid. The energy management system proposed in [26] represents decentralized control techniques used for the bipolar DC microgrid. The proposed system controls power sharing between individual parts of the microgrid and utility grid according to the load demand. In addition to power control, paper [26] proposed voltage control which will be discussed in Section 4. In paper [27], power management with voltage control was introduced. Power management was implemented to distribute available power to a fleet of EVs used as taxis. The proposed power management is capable of modifying the amount of energy sent to EVs depending on the microgrid situation. Paper [28] introduces intelligent hybrid energy management control to provide a reliable power balance for microgrid operation. The proposed control aims to effectively address fluctuations in a microgrid and enable reliable EV charging.

Only two of the reviewed papers with power flow control define their proposed control algorithms for AC microgrids. Authors in [29] study power quality improvement or precisely the active and reactive power control. They have proposed an approach to optimally control the active and reactive power of ultra-fast charging stations (UFCS) and RESs. Control strategy aims to coordinate the demands of EVs and other loads with

9 of 25

power provided by RESs and to minimize the charging time of connected EVs. Reactive power regulation was achieved by control of UFCS and PV converters. Paper [30] presents a P/V droop voltage controller for voltage regulation by controlling active power flow between EVs and microgrids. The proposed controller ensures utilization of EVs as a distributed energy storage system. Through the utilization of this controller, EVs were used to provide peak shaving and valley filling, thus contributing to the stable operation of the islanded microgrid.

More studies proposed their algorithms for hybrid microgrids. A power management strategy is introduced in [31] for a hybrid microgrid with a hybrid energy storage system to enhance power sharing among subgrids of hybrid microgrids and improve steady state and transient response, as well as to enhance the voltage stability of both subgrids. To test the effectiveness of the proposed control strategy, authors performed simulations for varying loads, PV power outputs, DG outages, and EV charging. Similarly, in [32] the power flow control scheme was introduced to improve power stability, power quality, system reliability, and optimal power distribution in hybrid microgrids. The proposed control provided continuous and reliable power to all loads connected to the microgrid as well as to the CS. The control scheme in [32] was based on an adaptive neural network Q-learning full recurrent adaptive neuro-fuzzy control. Paper [33] presents a new configuration for a unified power quality controller (UPQC) in combination with EVCS to improve power quality while ensuring the demand of EVCS. UPQC control is modified to address the charging of EVs by using power from the utility grid directly while compensating for power quality issues. During power outages or power interruptions, EVs were used through UPQC control as emergency power sources. Energy management for optimal power flow from the PV system and the utility grid to EVCS, BSS, and loads is described in [34]. The proposed control is used to optimize the EVCS load for Aligarh Muslim University while reducing operational costs of the EVCS microgrid.

There are also a few papers that do not define the information about the proposed microgrid architecture, such as in [35–38]. In Parmar's work [35], a fuzzy logic-based control is proposed to control the charging and discharging of plug-in-hybrid EVs as well as to manage other loads connected to the microgrid. Fuzzy logic control implements the V2G concept to allow the utilization of EVs with different charging profiles to provide support to the microgrid for optimum management of the daily load demand. Paper [36] presents a demand management approach to reduce peak demand by utilizing power from RES and EVCS. The proposed system consists of residential loads, critical commercial loads, RESs, and EVCS. The EVCS is represented by BSS and EVs, which are used as charging/discharging loads. The proposed method is capable of reducing peak demand, increasing the reliability of the microgrid, and increasing the profits of EVCS. Paper [37] presents a Hunger Games Search optimization algorithm to optimally allocate distributed generators into the microgrid with the aim to reduce power losses in the presence of an EVCS. In Olama's work [38], a Lyapunov-based hybrid model predictive control is proposed for energy management of renewable energy sources, different ESS, and EV's batteries to achieve safe and reliable operation of the microgrid. The proposed control uses EV's batteries to avoid fluctuations in energy demand to achieve optimal operation. The proposed energy management strategy is capable of dealing with energy management problems as well as ensuring the close-loop stability of the microgrid.

Information about the microgrid systems of the papers mentioned in this section is presented in Table 1.

ЪĆ	Ν	/licrogrid	Energy G	eneration	Enorou Storago	Other Loads Char Station	
Kef.	Туре	Architecture	RES	CES	Energy Storage	Other Loads	Char. Stations
[17]	DC	Islanded	PV WT	n.p.	Battery storage	\checkmark	DC FSC
[18]	DC	Grid connected	PV	n.p.	Battery storage	\checkmark	n.d.
[19]	DC	Grid connected	PV WT FC	n.p.	Battery storage Hydrogen storage	×	FCS
[20]	DC	Grid connected	PV	n.p.	Battery storage	×	FCS
[21]	DC	Grid connected	PV	n.p.	Battery storage	×	DC
[22]	DC	Grid connected	PV	n.p.	Battery storage	×	DC
[23]	DC	Grid connected	PV SHG	n.p.	Battery storage	×	DC
[24]	DC	Grid connected	PV	n.p.	Battery storage	×	FCS
[25]	DC	Grid connected	PV	n.p.	Battery storage	×	DC
[26]	DC	Islanded	PV	n.p.	Battery storage	\checkmark	DC
[27]	DC	Grid connected	PV WT	n.p.	Battery storage	×	FCS DC
[27]	DC	Islanded	PV WT FC	n.p.	Battery storage	\checkmark	n.d.
[28]	AC	Grid connected	PV	n.p.	n.p.	\checkmark	UFCS
[30]	AC	Islanded	PV WT	DG	n.p.	×	AC
[31]	Hybrid	Grid connected	PV	DG	Battery storage Supercapacitor FCES	×	n.d.
[32]	Hybrid	Grid connected	PV WT FC	MT	Battery storage Supercapacitor Hydrogen storage	\checkmark	AC
[33]	Hybrid	Grid connected	n.p.	n.p.	n.p.	\checkmark	DC
[34]	Hybrid	Grid connected	PV	n.p.	Battery storage	\checkmark	AC
[35]	n.d.	Islanded	PV WT	DG	n.p.	\checkmark	n.d.
[36]	n.d.	Grid connected	PV	n.p.	Battery storage	\checkmark	n.d.
[37]	n.d.	Grid connected	n.d.	n.p.	n.p.	\checkmark	AC
[38]	n.d.	Grid connected	PV WT FC	DG	Battery storage Supercapacitor Hydrogen storage	\checkmark	AC

Table 1. Information about the microgrid systems with power flow control.

3.2. Controlling the Self-Consumption of Microgrids

Paper [39] presents a novel DC microgrid power architecture for efficient charging of EVs. The proposed microgrid is based on PV generation, BSS, and grid connection, with EVs having direct access to their DC charger input. Unlike conventional power architecture designs, the PV system is directly coupled to the DC link without a static converter, which increases energy efficiency and reduces control complexity. The power management proposed in [39] controls power output from a PV system, which thereby enables self-consumption of the energy produced within microgrid and the utility grid is used only as a backup power source. The authors in [40] proposed an energy management system based on hysteresis V2G control. The designed system is capable of minimizing the impact of continuous charging current modulation on the life of EV batteries and of maximizing microgrid self-consumption. The authors simulated the proposed system with data measured in an actual microgrid with intermittent RESs production. Papers [41–43] present the implementation and evaluation of self-consumption operational strategies based on data from a microgrid at the EUREF-Campus in Berlin, Germany. The focus of the operational strategy in [41,42] is on supplying energy for EVCSs by controlling the energy generation of local sources and energy supply from the utility grid. The presented operational strategy is capable of achieving the sustainable application of EVs in an urban microgrid and with the deployment of a BSS can achieve CO_2 emissions reduction. In Feizi's work [43], an optimalization algorithm for demand side management is used to reduce load peaks by using the EVs and BSS as flexible loads. The presented algorithm was able to achieve 100% self-sufficiency.

Information about the microgrid systems of the papers mentioned in this section is presented in Table 2.

D (I	Microgrid	Energy G	Generation	Enorou Storago		
Kef.	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations
[39]	DC	Grid connected	PV	n.p.	Battery storage	×	DC FSC
[40]	AC	Grid connected	PV	n.p.	n.p.	\checkmark	AC
[41]	AC	Grid connected	PV	CHP	Battery storage	×	AC
[42]	AC	Grid connected	PV	CHP	Battery storage	×	AC
[43]	AC	Grid connected	PV	n.p.	Battery storage	×	AC

 Table 2. Information about the microgrid systems with self-consumption control.

3.3. Controlling Maximum Utilization of RES and Power Quality in Microgrids

The maximum use of RESs within a microgrid provides the possibility to decrease the microgrid dependency on the utility grid. The paper in [44] discusses the technical issues related to the uncoordinated charging of EVs and how this can lead to ineffective utilization of RESs connected to the CS and the microgrid, and how this can cause overloading of the utility grid. For this reason, the authors of [44] proposed an energy management strategy with aims to minimize the usage of utility grid power and to maximize the use of PV power as well as to store PV power when the CS is not used. The proposed energy management also controls the charging of EVs and BSS and based on the state of charge of the EVs and BSS determines the operational mode of microgrid operation. In Locment's work [45], a control strategy for maximizing power extraction from the PV sources and power flow management is designed to consider the state of charge of the EVs and power demand of the load connected to the DC microgrid. The maximum use of RESs can also be used to improve the social welfare of EVCS users as is demonstrated in [46]. The authors of the paper used power from a PV system to charge EVs which led to the maximalization of social welfare. This charging control was introduced to optimally distribute available power from a PV system to connected EVs. Paper [47] proposes a control strategy for microgrids with EVs. The proposed control strategy is employing a multi-objective technique that aims to achieve power factor regulation as well as economic battery charging. The control strategy uses the EVCS control to support power factor issues, allowing for full PV generation.

Information about the microgrid systems of the papers mentioned in this section is presented in Table 3.

	Microgrid		Energy Generation		En ener Stevense		
Kef.	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations
[44]	Hybrid	Grid connected	PV	n.p.	Battery storage	\checkmark	DC
[45]	DC	Grid connected	PV WT	n.p.	Battery storage	×	FCS
[46]	AC	Grid connected	PV	n.p.	n.p.	×	AC
[47]	AC	Grid connected	PV	n.p.	n.p.	\checkmark	AC

Table 3. Information about the microgrid systems with maximum utilization of RES and power quality control.

4. Voltage Control Strategies of Microgrids

Another strategy commonly used in research papers for microgrid control is voltage control. Voltage control provides the possibility for reliable control of individual components of microgrids, or for control of microgrid stability, or to design power converters and controllers with specific control structures. The papers with voltage control objectives are discussed in this section.

4.1. Microgrid Voltage Sag Mitigation Control

Papers [48,49] present a decentralized control system for microgrids with EV fast charging stations. The aim of this control is to reduce utility grid reliance and improve power quality. To improve power quality, the authors are using a distribution static compensator (D-STATCOM) with the objective of mitigating the voltage sag within a microgrid. The authors in both papers proposed the utilization of an EV battery as a DC source of a D-STATCOM through the V2G technology. One of the differences between these two papers is the utilization of different control methods. In Mohammed's work [48], the authors used a sliding mode control together with a fuzzy logic control, while in [49] the authors used a decentralized control system based on Proportional Integral (PI) controllers.

Paper [50] presents a coordinated control strategy of dynamic voltage restorer (DVR) and EVCS in a microgrid. In this paper, two categories of voltage sag (interior, exterior) are addressed with different compensating methods. For interior voltage sag, only the DVR is used, while on the other hand for exterior sag, DVR with cooperation from EVCS is used.

In Table 4, information about the microgrid systems of the papers mentioned in this section is presented.

D (I	Microgrid	Energy G	Generation	En aver Stavesa		
Kef.	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations
[48]	DC	Grid connected	PV	n.p.	n.p.	\checkmark	FCS
[49]	DC	Grid connected	PV	DG	n.p.	×	FCS
[50]	AC	Grid connected	n.p.	n.p.	n.p.	×	n.d.

Table 4. Information about the microgrid systems with voltage sag mitigation control.

4.2. Controlling the Bus Voltage of Microgrids

Most of the studies with bus voltage control have focused on grid-connected microgrid architectures. In paper [51], a bus voltage control strategy for a hybrid microgrid with EVCS is proposed. The proposed bus voltage control is maintained by a novel uninterruptible inverter which is used to provide uninterrupted power to the EVCS as well as ensure stable operation of the microgrid. Paper [52] presents a control strategy based on the virtual battery model. The primary aim of this control strategy is to control the bus voltage, dispatch power among the individual elements of the microgrid, and reduce the negative impact of EV fast charging on the electrical grid. In García-Triviño's work [20], the authors

proposed a decentralized control method with fuzzy logic controllers to coordinate power flow as was mentioned in Section 3.1 with bus voltage and battery energy storage state of charge (BSS SOC). The bus voltage control was used to ensure power balance among the components within a microgrid (EV FCS, PV system, BSS, utility grid) and to control the BSS SOC. Similarly, as with [20] a decentralized control was proposed in paper [53], but in this paper a PI controller was used for the bus voltage control. DC bus voltage in [53] was controlled by the power sources within the microgrid to ensure stable charging of EVs. Paper [54] proposes smart charging of PEV with a small scale wind energy system as a source. The control strategy proposed in [54] is the coordination control based on DC link voltage to support the operation of EVCS in stand-alone as well as in grid-connected mode. To provide an efficient power management of a DC microgrid, a power reference-based drop controller is designed to regulate bus voltage through energy storage. The intelligent control method for DC FCS is proposed in [55]. The authors are using the comprehensive AC/DC converter control to inject reactive power into a network to control the voltage and thus keep DC bus voltage at a constant value.

On the other hand, there are just a few studies that have dealt with the islanded operation of the microgrids. The authors in [56] proposed an enhanced control method for UFCS in islanded DC microgrids to reduce the transients of the DC bus voltage. To achieve this, the authors proposed an enhanced control strategy for the local controller of the EV converter. The hierarchical control strategy for a PV-based microgrid with EV wireless charging station is presented in [57]. The control structure is composed of central and local controllers where the central controller is used to assess the operation mode of the system and to select the local controllers and the local controller is used for battery storage control, high frequency (HF) inverter DC bus voltage control, and EV charging control. The control structure proposed in [57] can through HF inverter DC voltage controller regulate DC bus voltage and thus change the wireless charging power. In Bhargavi's work [58], a control strategy for EV is proposed to increase the maximum rate of charging capacity only if generation exceeds the local demand.

Table 5 presents information about the microgrid systems with bus voltage control discussed in this section.

	Ν	Aicrogrid	Energy G	eneration	En aver Stavage		
Ref.	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations
[51]	Hybrid	Grid connected	PV	n.p.	Battery storage	\checkmark	AC
[52]	DC	Grid connected	PV	n.p.	Battery storage	×	FCS
[53]	DC	Grid connected	PV	n.p.	Battery storage	×	FCS
[54]	DC	Grid connected	PV WT	n.p.	Battery storage	×	DC
[55]	DC	Grid connected	n.p.	n.p.	n.p.	×	FCS
[56]	DC	Islanded	PV	n.p.	Battery storage	×	UFCS
[57]	DC	Islanded	PV	n.p.	Battery storage	\checkmark	Inductive
[58]	DC	Islanded	PV WT	n.p.	Battery storage	\checkmark	DC

Table 5. Information about the microgrid systems with bus voltage control.

4.3. Voltage Regulation and Stability Control

In paper [59], a controller for an EVCS DC microgrid is designed to keep voltage stability and suppress the power system dynamics. To design the DC microgrid controller, the authors proposed a firefly algorithm combined with particle swarm optimalization. Paper [60] describes a new approach for the integration of EVs into the electrical grid, with the aim to reduce their impact on the grid and to provide ancillary services to the AC utility

grid. The proposed control can stabilize the voltage of a DC microgrid while providing ancillary services. Additionally, the authors demonstrated that use of V2G mode in microgrids with EVCS can help to support the stability of the electrical system. The adaptive bidirectional droop control for an EV parking lot with V2G service is proposed in [61]. This control strategy is coupled with solid state transformer for the voltage regulation within the microgrid. To validate the effectiveness of the proposed control strategy, the authors conducted experimental testing using a physical prototype.

A close-loop control structure based on feedback signals from EVCS, distributed generators, and load demand in the microgrid is presented in [62] to increase the reliability and stability of the system. In paper [27], DC voltage control was implemented through a PI controller in the battery storage to manage the microgrid voltage. Additionally, power management was introduced in this paper as was discussed in Section 3.1. In Benamar's work [63], a nonlinear control was presented to ensure the voltage stability of a DC microgrid by considering the fluctuations of the PV production as well as the EV demand.

Information about the microgrid systems with voltage regulation and stability control discussed in this section is presented in Table 6.

D (Ν	/licrogrid	Energy G	eneration	En anon Standard			
Kef.	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations	
[59]	DC	n.d.	PV WT FC	n.p.	Battery storage Hydrogen storage	×	n.d.	
[60]	DC	Grid connected	PV	n.p.	n.p.	×	DC	
[61]	AC	Grid connected	PV	n.p.	Battery storage	×	DC	
[62]	Hybrid	Grid connected	PV WT	n.p.	Battery storage	\checkmark	DC	
[63]	DC	Islanded	PV	n.p.	Battery storage Supercapacitor	×	DC	

Table 6. Information about the microgrid systems with voltage regulation and stability control.

5. Economic Operation Control of Microgrids

Economic operation control of microgrids presents the different aspects of microgrid control used to achieve desired outcomes. Desired outcomes can be beneficial for microgrid operators, users of microgrids, or both based on the control objectives. The most commonly occurring objectives used in economic operation control are discussed in this section.

5.1. Maximizing the Profits of Microgrids and Energy Trading

Paper [64] presents a control solution for a smart microgrid with an EVCS to maximize the profits of microgrid owners by selling energy to EVs. To achieve this, the authors proposed the energy management system which controls the power profile of the microgrid. The coordinated dispatch method for PV power generation microgrids with EVs is proposed in [65]. The proposed method is using the power forecasting model based on real data to maximize the profits of microgrids and thus maximize the PV power consumption.

The paper in [66] introduces a predictive control algorithm for economic optimization in a microgrid that is connected to the utility grid and EVCS. The system is modeled using the Energy Hubs methodology, and the proposed algorithm is responsible for managing electricity purchase and sale to the grid. In Mendes's work [67], a control strategy for optimal operation of microgrids as well as energy sale and purchase into microgrids is introduced. The proposed strategy can manage properly the purchase and sale of energy to the external network, ensure the charging of EVs, and manage load demand. Multi-period optimal energy scheduling and trading for multi-microgrids is discussed in [68]. In this paper, comprehensive models of multi-microgrids integrated with an urban transportation network through fast charging stations is formulated. Trading schemes are used to maximize profits for each microgrid through buy-low-sell-high fashion and at the same time minimize travel expenses in response to the FCS charging prices defined by the individual microgrids.

Table 7 presents information about the microgrid systems with profit maximization and energy trading discussed in this section.

D (I	Microgrid	Energy C	Generation	En aver Channes		
Kef	Туре	Architecture	RES	CES	— Energy Storage	Other Loads	Char. Stations
[64]	AC	Grid connected	PV WT	n.p.	Battery storage	\checkmark	AC
[65]	n.d.	Grid connected	PV	n.p.	n.p.	×	n.d.
[66]	n.d.	Grid connected	PV WT	n.p.	Battery storage Hydrogen storage	\checkmark	n.d.
[67]	n.d.	Grid connected	PV	n.p.	Battery storage Hydrogen storage	×	FCS DC
[68]	n.d.	Grid connected	PV WT	DG	n.p.	\checkmark	FCS

Table 7. Information about the microgrid systems with profit maximization and energy trading.

5.2. Minimizing the Operational Costs of Microgrids and Electricity Costs

Paper [69] proposes a two-stage optimal framework to handle the online dispatch problem of grid connected microgrids. The first stage in the proposed framework is to minimize the total operating cost and the second stage determines the power allocation for CS. The proposed framework can effectively realize the optimal dispatch and reduce the operating cost. An optimal management system to minimize operational costs of hybrid microgrids is proposed in [70]. The proposed approach manages power sources connected to the microgrid together with EVs charging/discharging to reduce operational costs of microgrids. In Han's work [71], an optimal control method for microgrid systems with residential loads based on EV virtual energy storage is proposed. For optimal control, multi-objective optimization is used with minimal electricity cost, EV battery state of health, and carbon emissions as objective functions. Paper [72] presents economical scheduling of EVs connected to the microgrid to reduce electricity cost. To minimize the electricity cost, the authors used charging/discharging control of EVs connected to microgrids. An optimal scheduling strategy is presented in [73] to minimize operational costs of EV chargingswapping-storage integrated stations and microgrids. The proposed coordination of the charging process is capable to transfer charging demand from the peak demand time to valley demand time, thus reducing the peak-to-valley difference and reducing operational costs of microgrids. In Xu's work [74], similarly to previous papers, optimal scheduling is proposed. The aim in [74] is to reduce charging costs of EVs as well as to reduce power loss of microgrids to improve overall efficiency of microgrids. Paper [75] presents a real-time power management strategy for intelligent infrastructure for recharging electric vehicles based on a microgrid system. The strategy proposed in the paper aims to minimize the energy cost of microgrids and maximize the use of PV power. An energy management strategy and an optimal operation strategy of EVs and battery swapping stations are discussed in [14]. The objectives of the operational strategy are the minimization of operational costs and the maximization of benefits of microgrids in islanded operation mode. To minimize operational costs, authors adopted price mechanisms to flatten the demand curves which reduces the difference between the peak and valley demand. The optimal operation of a community based microgrid is presented in [76] to minimize operational costs of microgrids and thus maximizing economic benefits for the prosumers. The goal of the paper

is achieved by optimal scheduling of charging and discharging of EVs based on dynamic day ahead prices. Paper [77] presents an energy management system based on a dynamic optimalization model to minimize operational costs of microgrids as well as to minimize CO_2 emissions. The proposed approach is tested and applied to the case study of the Sanova University Campus. A day ahead demand schedule strategy is proposed in [15] to improve the cost-effectiveness of the microgrid. The proposed strategy is based on optimal load shifting of demand with incentive and penalty pricing as a decision support for uses. An energy management model for a smart community microgrid including battery swapping stations and residential loads is formulated in [78]. The optimization algorithm is formulated to minimize the average long-term energy cost of consumption and operational cost while securing the quality of service for battery swapping stations and residential loads. In Liao's work [79], an energy management system for a smart home and an EV charging parking lot is presented. Energy management aims to minimize electricity payment for smart home while ensuring users' comfort and minimizing costs of EV charging within the EV charging parking lot. The control strategy for smart homes involves control and scheduling of home appliances and scheduling is also used in the case of the EV parking lot. In the case of the EV parking lot, the EVs also join in an ancillary service market of emergency reserve capacity. Paper [80] presents real-time management and scheduling of EVs in a commercial microgrid with PV generation, partially curtailable load, and battery storage unit for backup. The proposed control structure is capable to allocate and distribute power based on the energy price thus minimizing the cost of power to the microgrid.

Information about the microgrid systems of the papers mentioned in this section is presented in Table 8.

D (licrogrid	Energy G	eneration	E. Classes		
Ket.	Туре	Architecture	RES	CES	 Energy Storage 	Other Loads	Char. Stations
[69]	DC	Grid connected	PV WT	n.p.	Battery storage	\checkmark	DC
[70]	Hybrid	Grid connected Islanded	PV FC	DG MT	n.p.	\checkmark	n.d.
[71]	AC	Grid connected	PV	n.p.	Battery storage	\checkmark	AC
[72]	DC	Grid connected	PV FC	MT	Battery storage	\checkmark	DC
[73]	n.d.	Grid connected	PV WT	MT	Battery storage	\checkmark	FCS Swapping
[74]	AC	Grid connected	PV	n.p.	Battery storage	\checkmark	AC
[75]	DC	Grid connected	PV	n.p.	Battery storage	×	FCS
[14]	AC	Grid connected Islanded	PV WT	n.p.	n.p.	\checkmark	AC Swapping
[76]	AC	Grid connected	PV	n.p.	Battery storage	\checkmark	AC
[77]	n.d.	Grid connected	PV WT	MT	Battery storage	\checkmark	n.d.
[15]	AC	Grid connected Islanded	PV WT FC	DG MT	Battery storage	\checkmark	AC
[78]	n.d.	Grid connected	PV WT	n.p.	n.p.	\checkmark	Swapping
[79]	AC	Grid connected	PV	n.p.	n.p.	✓	AC
[80]	AC	Grid connected	PV	n.p.	Battery storage	\checkmark	AC

Table 8. Information about the microgrid systems with objectives to minimize operational costs of microgrids and electricity costs.

6. Other Control Strategies

The control strategies discussed in the previous section represented the most commonly used control strategies for microgrid control. This section contains control strategies that were presented only in a few papers from our final set. The following subsections discuss combined control strategies for microgrid control in which the authors used multiple distinctive objectives to achieve desired outcomes and frequency controls.

6.1. Combine Control Strategy for Microgrid Control

Multi-microgrid network energy management is proposed in [81] to maximize the energy exchange between microgrids to reduce the load fluctuation, minimize the amount of energy purchased from the distribution network, and encourage self-healing capability to maintain the system operation during fault events. To help deal with disturbances and maintain system stability, the V2G concept is used during fault events. Paper [82] presents a control problem of microgrids integrating renewable generation, hybrid storage technologies, and interaction with V2G systems. The objectives of microgrid control were to maximize the use of renewable energy sources, coordinate the use of battery bank and hydrogen storage to minimize the oscillation between production and demand, perform the charging of EVs, and make the purchase and sale management of electricity to the external network. In paper [83], the authors suggest an optimal planning technique for identifying the locations and sizes of EVCS and hybrid RESs inside a microgrid. The main objectives of the proposed technique are minimization of voltage deviation, power loss, the cost of building renewable resources, and the cost of charging EVs. To solve the proposed problem, the authors used a novel jellyfish search optimizer.

Information about the microgrid systems of the papers mentioned in this section is presented in Table 9.

Table 9. Information about the microgrid systems with combined control strategies for microgrid control.

	I	Microgrid		eneration		0.1 1 1	
Ref.	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations
[81]	DC	Grid connected	PV WT	n.p.	Battery storage	\checkmark	FCS DC
[82]	DC	Grid connected	PV WT FC	n.p.	Battery storage Hydrogen storage	\checkmark	FCS DC
[83]	n.d.	Grid connected	PV WT	n.p.	n.p.	×	n.d.

6.2. Frequency Control of Microgrids

Paper [84] proposes a V2G strategy for primary frequency control inside an industrial microgrid. The primary frequency control is achieved by the charging and discharging of EVs based on the signals from the industrial grid operator. Paper [85] presents the droop-controlled charging/discharging strategy of EVs to ensure frequency stability in an islanded microgrid through primary frequency control. The proposed control strategy considers EV state of charge to maintain battery lifetime and ensures the availability of the EV as needed.

Table 10 presents information about the microgrid systems with frequency control discussed in this section.

	Microgrid		Energy Generation		Ere error Store ere		
Kef	Туре	Architecture	RES	CES	- Energy Storage	Other Loads	Char. Stations
[84]	n.d.	Grid connected	PV WT	n.p.	n.p.	×	AC
[84]	AC	Islanded	PV	DG	n.p.	\checkmark	AC

 Table 10. Information about the microgrid systems with frequency control of microgrids.

7. Discussion

Integration of renewable energy resources and electric vehicle charging stations into the microgrid system has emerged as a significant research subject due to the potential it presents for the efficient use of these resources. This integration offers the possibilities to increase the reliability and self-sufficiency of microgrids while enabling them to provide ancillary services to the utility grid. To achieve the desired operational objectives, researchers have designed and tested various control mechanisms, as highlighted in the reviewed literature. The most commonly implemented control mechanisms discussed in the literature are power control, voltage control, and economic operation control of microgrid systems, as indicated in Figure 8. Each of these control mechanisms has been discussed in detail in previous sections, with corresponding literature cited for further reference and understanding. These control strategies play a key role in optimizing the performance and overall efficiency of microgrid systems, thereby preparing the way for sustainable and resilient energy management designs.

7.1. Operational and Control Strategies

The main aim of this review article is to provide a comprehensive overview of the different operational strategies employed in microgrids that incorporate electric vehicle charging stations. As mentioned above, the classification of various operational strategies or objectives has been discussed in previous sections. In the literature review, the use of charging stations has been observed in two different forms: active participation through controlled charging/discharging and passive participation as uncontrolled charging where the charging stations are considered only as loads. Figure 9 illustrates the distribution of charging station utilization in microgrid operational control. Notably, of the total number of papers reviewed (70), 46 papers actively employed charging stations for controlled charging to achieve specific goals based on the microgrid control strategy adopted. This active use of charging stations highlights their importance in improving the operational efficiency and overall performance of microgrids while facilitating a range of control options to optimize energy management and grid stability.



Figure 9. Charging station operation distribution.

For successful implementation of control strategies in real-world applications, it is important to validate these strategies using models that combine real-world data from the RESs and charging stations operation, EV user behavior, load demand, and other components of the microgrid. However, in the reviewed literature, the most common approach among the authors was to test their proposed control strategies in microgrid simulations with simplified models of EV charging habits without using actual real-world data. This lack of inclusion of real-world data in the testing may introduce uncertainties and limit the accuracy of the results. In papers [34,43,71], the authors used real EV charging data for simplified estimation of charging station behavior, and in [77,78] the authors used information from travel surveys. The authors of these papers used real-world data for charging stations representation and charging stations are actively participating in proposed control strategies. Data about real charging station power consumption were also used in [52,74] to test the proposed control strategies, despite the fact that the charging station participated only passively in the proposed control strategies. To represent more realistic behavior of charging stations during the testing of a proposed control strategy, it is important to use real EVs charging data that can represent EVs charging behavior in a specific microgrid system more accurately. Through implementation of different methods of data generation, it is possible to generate new simulation input data for charging station behavior, and thereby simulate different sets of scenarios with the same historical data. The most used method in the reviewed literature is the utilization of probability distribution functions, which are based on real EV charging data and can generate new simulation input data based on the shape of the distribution function [15,24,25,62,73,80]. Other methods used in the reviewed literature to generate data based on real data include the Monte Carlo method ([24,65]) and the Markov method [76]. These methods are used to simulate and generate data that mimic real-world properties. The Monte Carlo method was also used in [14,72] to generate SOC parameters for EVs. In papers [24,65], the authors used real data for charging station representation despite the fact that charging stations acted only passively in their control strategies. To implement the proposed strategies, it is imperative to test them by using real-world data.

7.2. Interaction of Proposed Control Strategies with Real-World Applications

Simulated operation time is another important aspect connected to the assessment of results for the proposed control strategy and in assessing if the proposed control strategy can be implemented in real-world application. In the reviewed literature, thirty-three papers simulated their control strategies on 24 h simulation operation and only four papers ([24,69,78,80]) used a simulation time longer than 24 h. Longer simulation time can provide the possibility to test the proposed system with more complex behavior and represent more precisely the real-world application.

In papers [19,21,22,31,44,57], the authors compared simulations of their control strategies on experimental models to see how real components could react with the proposed control. Compering simulation results with results from an experimental model provides a more comprehensive understanding of how the proposed control strategy will function in real-world implementation.

7.3. Research Shortcomings and Research Gaps

The literature review presented in this paper highlights several research shortcomings that require further attention to ensure the successful implementation of the proposed control strategies in real microgrid systems. Further research is necessary to effectively address these shortcomings. Specifically, future research efforts should prioritize the use of real data to test and validate control strategies, especially in charging station modeling. By incorporating real data, the reliability and applicability of the proposed methodologies can be improved in real microgrid conditions. Another shortcoming that often appears in the reviewed literature is that the presented control strategies are simulated on short time periods, which allows for better presentation of the results but does not allow us to see the more complex behavior of the simulated system. On the other hand, longer simulation times allow us to evaluate the long-term performance and see the more complex behavior of the proposed control algorithms on simulated systems. This literature review also revealed some research gaps, notably the aspect of validation of the proposed controls strategies through comprehensive comparison between simulated results and results obtained from experimental testing or real-world implementation on a microgrid system. Such a comparison provides indisputable evidence of the functionality and effectiveness of the proposed control approach, but the length of the operation window and different operation states used for comparison are also of utmost importance. Another research gap is the absence of a control strategy that would address the preservation of reserved capacity in the microgrid. This could provide a new approach for the power exchange between microgrid and main grid as well as increase the economic benefits of microgrid operation.

8. Conclusions and Research Prospects

8.1. Conclusions

With the global shift towards sustainability and the rapid deployment of renewable energy resources, electric vehicles, and charging stations, it becomes crucial to explore strategies that can provide grid support while addressing the inherent intermittent nature of renewables and EVs. In this context, microgrids offer an ideal solution as they can operate independently or in coordination with the main grid, facilitating the generation, distribution, and management of power flow among interconnected resources.

The growing number of EVs can significantly impact the operation of power systems, requiring more sophisticated control strategies to match EV charging patterns with grid conditions. This close interplay between electrical power systems and the transportation sector underscores the importance for advanced control measures to ensure the seamless integration of EVs into microgrids, thereby safeguarding a stable and resilient power supply for consumers of EV charging stations as well as stable and efficient operation of microgrids. In response to these challenges, researchers have been developing and testing various control strategies for microgrids with EV charging stations in recent years.

8.2. Research Prospects

This review paper aims to offer a comprehensive overview of the different control strategies proposed in the literature to control microgrids with electric vehicle charging stations. Additionally, this paper sought to identify common shortcomings and research gaps in the current literature. Among the most common shortcomings identified in the existing literature are the limited number of publications utilizing real EV charging data, the simplistic approach in determining EV charging station consumption, and the relatively short simulation time periods for the proposed control strategies. Addressing these shortcomings in future control strategies is crucial to increase the effectiveness and robustness of the proposed approaches and to increase their real-world applicability. In addition, this review highlighted gaps in current research, including the limited number of control strategies that have been tested on experimental microgrid models or directly in real microgrid settings. Notably, a significant research gap in the current literature is the absence of control strategies specifically adapted to address the preservation of reserved capacity in microgrids using EV charging stations.

To address these gaps and to advance microgrid control strategies, future research should focus on incorporating real EV charging data, adopting more sophisticated models to represent realistic EV charging station behavior, and conducting tests with longer simulation times to evaluate the long-term performance of control strategies. The experimental validation of the proposed control strategies within microgrid configurations, along with the development of novel control strategies addressing the preservation of reserved capacity of microgrids will further contribute to significantly improve the overall efficiency and practical implementations of microgrid systems with EV charging stations.

Furthermore, to facilitate a comprehensive comparison and evaluation of different control strategies in both current and future research works, it is essential to establish a universal benchmark network. This benchmark network will serve as a standardized plat-

form for testing and evaluating the effectiveness of the proposed control strategies, thereby providing a deeper understanding of their application in different microgrid scenarios.

Author Contributions: Conceptualization, M.T.; methodology, M.T.; validation, M.T and M.K.; formal analysis, M.T.; investigation, M.T.; resources, M.T.; data curation, M.T.; writing-original draft preparation, M.T.; writing-review and editing, M.T., M.K. and P.B.; visualization, M.T., M.K.; supervision, P.B. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

Nomenclature

AC	Alternating Current
BSS	Battery Storage System
CES	Conventional Energy Sources
CS	Charging Station
DC	Direct Current
DG	Diesel Generator
D-STATCOM	Distribution Static Compensator
DVR	Dynamic Voltage Restorer
ESS	Energy Storage System
EV	Electric Vehicle
EVCS	Electric Vehicle Charging Station
FC	Fuel Cells
FCES	Fuel Cell Energy Storage
FCS	Fast Charging Station
HF	High Frequency
HSS	Hydrogen Storage Systems
CHP	Combined Heat and Power
MT	Microturbine
n.d.	not defined
n.p.	not present
PI	Proportional Integral
PV	Photovoltaic
RES	Renewable Energy Source
SC	Supercapacitor
SHG	Small Hydro Generators
SOC	State of Charge
UFCS	Ultra-Fast Charging Station
UPQC	Unified Power Quality Controller
V2G	Vehicle-to-Grid
V2V	Vehicle-to-Vehicle
WT	Wind Turbine

References

- 1. Colak, I. Introduction to smart grid. In Proceedings of the 2016 International Smart Grid Workshop and Certificate Program (ISGWCP), Istanbul, Turkey, 21–25 March 2016; pp. 1–5. [CrossRef]
- Danley, D.R. Defining a Microgrid Using IEEE 2030.7. Business & Technology Surveillance. 2019. Available online: https://www.cooperative.com/programs-services/bts/Documents/TechSurveillance/Surveillance-Defining-Microgrids-November-2019.pdf (accessed on 16 June 2023).
- 3. Rezaeimozafar, M.; Amini, M.H.; Moradi, M.H. Innovative appraisement of smart grid operation considering large-scale integration of electric vehicles enabling V2G and G2V systems. *Electr. Power Syst. Res.* **2018**, *154*, 245–256. [CrossRef]
- Savio Abraham, D.; Verma, R.; Kanagaraj, L.; Giri Thulasi Raman, S.R.; Rajamanickam, N.; Chokkalingam, B.; Marimuthu Sekar, K.; Mihet-Popa, L. Electric Vehicles Charging Stations' Architectures, Criteria, Power Converters, and Control Strategies in Microgrids. *Electronics* 2021, 10, 1895. [CrossRef]
- 5. Nimalsiri, N.I.; Mediwaththe, C.P.; Ratnam, E.L.; Shaw, M.; Smith, D.B.; Halgamuge, S.K. A Survey of Algorithms for Distributed Charging Control of Electric Vehicles in Smart Grid. *IEEE Trans. Intell. Transp. Syst.* **2020**, *21*, 4497–4515. [CrossRef]

- 6. Wu, Y.; Wang, Z.; Huangfu, Y.; Ravey, A.; Chrenko, D.; Gao, F. Hierarchical Operation of Electric Vehicle Charging Station in Smart Grid Integration Applications—An Overview. *Int. J. Electr. Power Energy Syst.* **2022**, 139, 2022. [CrossRef]
- Casella, V.; Fernandez Valderrama, D.; Ferro, G.; Minciardi, R.; Paolucci, M.; Parodi, L.; Robba, M. Towards the Integration of Sustainable Transportation and Smart Grids: A Review on Electric Vehicles' Management. *Energies* 2022, 15, 4020. [CrossRef]
- Hussain, M.T.; Sulaiman, N.B.; Hussain, M.S.; Jabir, M. Optimal Management Strategies to Solve Issues of Grid Having Electric Vehicles (EV): A Review. J. Energy Storage 2021, 33, 102114. [CrossRef]
- 9. Vadi, S.; Padmanaban, S.; Bayindir, R.; Blaabjerg, F.; Mihet-Popa, L. A Review on Optimization and Control Methods Used to Provide Transient Stability in Microgrids. *Energies* **2019**, *12*, 3582. [CrossRef]
- Erick, A.O.; Folly, K.A. Reinforcement Learning Approaches to Power Management in Grid-tied Microgrids: A Review. In Proceedings of the 2020 Clemson University Power Systems Conference (PSC), Clemson, SC, USA, 10–13 March 2020; pp. 1–6. [CrossRef]
- Kaur, S.; Kaur, T.; Khanna, R.; Singh, P. A State of the Art of DC Microgrids for Electric Vehicle Charging. In Proceedings of the 2017 4th International Conference on Signal Processing, Computing and Control (ISPCC), Solan, India, 21–23 September 2017; pp. 381–386. [CrossRef]
- 12. Hu, J.; Morais, H.; Sousa, T.; Lind, M. Electric vehicle fleet management in smart grids: A review of services, optimization and control aspects. *Renew. Sustain. Energy Rev.* 2016, 56, 1207–1226. [CrossRef]
- Alam, M.S.; Arefifar, S.A. Energy Management in Power Distribution Systems: Review, Classification, Limitations and Challenges. IEEE Access 2019, 7, 92979–93001. [CrossRef]
- Zhang, M.; Chen, J. The Energy Management and Optimized Operation of Electric Vehicles Based on Microgrid. *IEEE Trans.* Power Deliv. 2014, 29, 1427–1435. [CrossRef]
- 15. Battula, A.R.; Vuddanti, S.; Salkuti, S.R. A Day Ahead Demand Schedule Strategy for Optimal Operation of Microgrid with Uncertainty. *Smart Cities* 2023, *6*, 491–509. [CrossRef]
- IEA. Global EV Outlook 2023, IEA, Paris. Available online: https://www.iea.org/reports/global-ev-outlook-2023 (accessed on 14 June 2023).
- 17. Sayed, K.; Abo-Khalil, A.G.S.; Alghamdi, A. Optimum Resilient Operation and Control DC Microgrid Based Electric Vehicles Charging Station Powered by Renewable Energy Sources. *Energies* **2019**, *12*, 4240. [CrossRef]
- Hadero, M.; Khan, B. Development of DC Microgrid Integrated Electric Vehicle Charging Station With Fuzzy Logic Controller. Front. Energy Res. 2022, 10, 922384. [CrossRef]
- García-Triviño, P.; de Oliveira-Assís, L.; Soares-Ramos, E.P.P.; Sarrias-Mena, R.; García-Vázquez, C.A.; Fernández-Ramírez, L.M. Supervisory Control System for a Grid-Connected MVDC Microgrid Based on Z-Source Converters With PV, Battery Storage, Green Hydrogen System and Charging Station of Electric Vehicles. *IEEE Trans. Ind. Appl.* 2023, 59, 2650–2660. [CrossRef]
- García-Triviño, P.; Torreglosa, J.P.; Fernández-Ramírez, L.M.; Jurado, F. Decentralized Fuzzy Logic Control of Microgrid for Electric Vehicle Charging Station. *IEEE J. Emerg. Sel. Top. Power Electron.* 2018, 6, 726–737. [CrossRef]
- 21. Abraham, D.S.; Balaji, C.; Kodandapani, D.; Sathyasekar, K.; Naryanmoorthi, R.; Bharatiraja, C.; Twala, B. DC Microgrid Integrated Electric Vehicle Charging Station Scheduling Optimization. *J. Appl. Sci. Eng.* **2022**, *26*, 253–260. [CrossRef]
- Abraham, D.S.; Chandrasekar, B.; Rajamanickam, N.; Vishnuram, P.; Ramakrishnan, V.; Bajaj, M.; Piecha, M.; Blazek, V.; Prokop, L. Fuzzy-Based Efficient Control of DC Microgrid Configuration for PV-Energized EV Charging Station. *Energies* 2023, 16, 2753. [CrossRef]
- Naik, K.R.; Rajpathak, B.; Mitra, A.; Kolhe, M. Renewable Energy Integrated DC Microgrid for EV Charging Station. In Proceedings of the 2021 IEEE Transportation Electrification Conference (ITEC-India), New Delhi, India, 16–19 December 2021; pp. 1–6. [CrossRef]
- 24. Engelhardt, J.; Zepter, J.M.; Gabderakhmanova, T.; Marinelli, M. Energy Management of a Multi-Battery System for Renewable-Based High Power EV Charging. *eTransportation* **2022**, *14*, 100198. [CrossRef]
- Dicorato, M.; Forte, G.; Trovato, M.; Muñoz, C.B.; Coppola, G. An Integrated DC Microgrid Solution for Electric Vehicle Fleet Management. *IEEE Trans. Ind. Appl.* 2019, 55, 7347–7355. [CrossRef]
- Nisha, K.S.; Gaonkar, N.D.; Jayalakshmi, N.S. Operation and Control of Multiple Electric Vehicle Load Profiles in Bipolar Microgrid with Photovoltaic and Battery Energy Systems. J. Energy Storage 2023, 57, 106261. [CrossRef]
- Noriega, B.E.; Pinto, R.T.; Bauer, P. Sustainable DC-microgrid Control System for Electric-Vehicle Charging Stations. In Proceedings of the 15th European Conference on Power Electronics and Applications (EPE), Lille, France, 2–6 September 2013; pp. 1–10. [CrossRef]
- Mohan, H.M.; Dash, S.K. Renewable Energy-Based DC Microgrid with Hybrid Energy Management System Supporting Electric Vehicle Charging System. Systems 2023, 11, 273. [CrossRef]
- 29. di Noia, L.P.; Mottola, F.; Proto, D.; Rizzo, R. Real Time Scheduling of a Microgrid Equipped with Ultra-Fast Charging Stations. Energies 2022, 15, 816. [CrossRef]
- Singh, S.; Jagota, S.; Singh, M. Energy Management and Voltage Stabilization in an Islanded Microgrid through an Electric Vehicle Charging Station. *Sustain. Cities Soc.* 2018, 41, 679–694. [CrossRef]
- Jithin, S.; Rajeev, T. Novel Adaptive Power Management Strategy for Hybrid AC/DC Microgrids with Hybrid Energy Storage Systems. J. Power Electron. 2022, 22, 2056–2068. [CrossRef]

- 32. Awais, M.; Khan, L.; Khan, S.G.; Awais, Q.; Jamil, M. Adaptive Neural Network Q-Learning-Based Full Recurrent Adaptive NeuroFuzzy Nonlinear Control Paradigms for Bidirectional-Interlinking Converter in a Grid-Connected Hybrid AC-DC Microgrid. *Energies* **2023**, *16*, 1902. [CrossRef]
- 33. Zhong, Y.; Xia, M.; Chiang, H.-D. Electric Vehicle Charging Station Microgrid Providing Unified Power Quality Conditioner Support to Local Power Distribution Networks. *Int. Trans. Electr. Energy Syst.* **2017**, 27, e2262. [CrossRef]
- Khan, S.; Ahmad, F.; Alam, M.S.; Krishnamurthy, M. Design and Optimization of Microgrid as EV Charging Source. In Proceedings of the Advances in Electrical and Computer Technologies, Bangkok, Thailand, 1–3 July 2020; Sengodan, T., Murugappan, M., Misra, S., Eds.; Lecture Notes in Electrical Engineering. Springer: Singapore, 2020; Volume 672. [CrossRef]
- Parmar, C.; Tiwari, S. Fuzzy Logic Based Charging of Electric Vehicles for Load Management of Microgrid. In Proceedings of the 5th International Conference on Communication and Electronics Systems (ICCES), Coimbatore, India, 15–16 December 2020; pp. 1–7. [CrossRef]
- 36. Hafeez, A.; Alammari, R.; Iqbal, A. Utilization of EV Charging Station in Demand Side Management Using Deep Learning Method. *IEEE Access* 2023, *11*, 8747–8760. [CrossRef]
- Eid, A.; Ibrahim, M.; Kamel, S. Optimal Planning of Microgrids Including Charging Stations and Renewable Energy Sources. In Proceedings of the 2021 22nd International Middle East Power Systems Conference (MEPCON), Assiut, Egypt, 14–16 December 2021; pp. 503–508. [CrossRef]
- Olama, A.; Mendes, P.R.C.; Camacho, E.F. Lyapunov-Based Hybrid Model Predictive Control for Energy Management of Microgrids. IET Gener. Transm. Distrib. 2018, 12, 5770–5780. [CrossRef]
- 39. Locment, F.; Sechilariu, M. Modeling and Simulation of DC Microgrids for Electric Vehicle Charging Stations. *Energies* 2015, *8*, 4335–4356. [CrossRef]
- Damiano, A.; Musio, C.; Sanna, C.; Gawronska, M. A Microgrid Energy Management System Supported by a Hysteresis Vehicle to Grid Controller. In Proceedings of the 2015 International Conference on Renewable Energy Research and Applications (ICRERA), Palermo, Italy, 22–25 November 2015; pp. 879–884. [CrossRef]
- 41. Khemir, M.; Rojas, M.; Popova, R.; Feizi, T.; Heinekamp, J.F.; Strunz, K. Real-World Application of Sustainable Mobility in Urban Microgrids. *IEEE Trans. Ind. Appl.* **2022**, *58*, 1396–1405. [CrossRef]
- Khemir, M.; Rojas, M.; Popova, R.; Feizi, T.; Nguyen, H.Q.; Heinekamp, J.F.; Strunz, K. Real-World Application of Sustainable Mobility in Urban Micro Smart Grids. In Proceedings of the 2020 International Conference on Smart Energy Systems and Technologies (SEST), Istanbul, Turkey, 7–9 September 2020; pp. 1–6. [CrossRef]
- Feizi, T.; von der Heiden, L.; Popova, R.; Rojas, M.; Gerbaulet, J. Day-Ahead Optimization Algorithm for Demand Side Management in Microgrids. In Proceedings of the 8th International Conference on Smart Cities and Green ICT Systems— SMARTGREENS, Crete, Greece, 3–5 May 2019; SciTePress: Setúbal, Portugal, 2019; pp. 51–57, ISBN 9789897583735. [CrossRef]
- 44. Savio, D.A.; Juliet, V.A.; Chokkalingam, B.; Padmanaban, S.; Holm-Nielsen, J.B.; Blaabjerg, F. Photovoltaic Integrated Hybrid Microgrid Structured Electric Vehicle Charging Station and Its Energy Management Approach. *Energies* **2019**, *12*, 168. [CrossRef]
- Locment, F.; Sechilariu, M. DC Microgrid for Future Electric Vehicle Charging Station Designed by Energetic Macroscopic Representation and Maximum Control Structure. In Proceedings of the 2014 IEEE International Energy Conference (ENERGYCON), Cavtat, Croatia, 13–16 May 2014; pp. 1454–1460. [CrossRef]
- 46. Kajanova, M.; Bracinik, P. Social welfare-based charging of electric vehicles in the microgrids fed by renewables. *Int. J. Electr. Power Energy Syst.* 2022, 138, 107974. [CrossRef]
- 47. Shahab, M.; Wang, S.; Junejo, A.K. Improved Control Strategy for Three-Phase Microgrid Management with Electric Vehicles Using Multi Objective Optimization Algorithm. *Energies* **2021**, *14*, 1146. [CrossRef]
- Mohammed, A.M.; Alalwan, S.N.H.; Taşcıkaraoğlu, A.; Catalão, J.P.S. Sliding Mode-Based Control of an Electric Vehicle Fast Charging Station in a DC Microgrid. Sustain. Energy Grids Netw. 2022, 32, 100820. [CrossRef]
- Alalwan, S.N.H.; Mohammed, A.M.; Taşcıkaraoğlu, A.; Catalão, J.P.S. An Improved Energy Management Strategy for a DC Microgrid including Electric Vehicle Fast Charging Stations. In Proceedings of the 2021 International Conference on Smart Energy Systems and Technologies (SEST), Vaasa, Finland, 6–8 September 2021; pp. 1–6. [CrossRef]
- Lei, Z.; Ai, X.; Cui, M. Coordinated Control Strategy Based on Network Parameters for Voltage Sags Compensating in Microgrid. In Proceedings of the 2010 5th International Conference on Critical Infrastructure (CRIS), Beijing, China, 20–22 September 2010; pp. 1–5. [CrossRef]
- Chen, Y.; Wei, W.; Zhang, F.; Liu, C.; Meng, C. Design of PV Hybrid DC/AC Microgrid for Electric Vehicle Charging Station. In Proceedings of the 2017 IEEE Transportation Electrification Conference and Expo, Asia-Pacific (ITEC Asia-Pacific), Harbin, China, 7–10 August 2017; pp. 1–6. [CrossRef]
- 52. Wang, H.; Lu, L.; Han, X.; Ouyang, M.; Feng, X. Virtual-battery based droop control and energy storage system size optimization of a DC microgrid for electric vehicle fast charging station. *Appl. Energy* **2020**, *259*, 114146. [CrossRef]
- García-Triviño, P.; Torreglosa, J.P.; Fernández-Ramírez, L.M.; Jurado, F. Control and Operation of Power Sources in a Medium-Voltage Direct-Current Microgrid for an Electric Vehicle Fast Charging Station with a Photovoltaic and a Battery Energy Storage System. *Energy* 2016, 115, 38–48. [CrossRef]
- Vijayakumar, R.; Malarvizhi, K.; Divyapriya, S. Design of Public Plug-in Electric Vehicle Charging Station for Improving LVRT Capability of Grid Connected Wind Power Generation. In Proceedings of the 2018 International Conference on Soft-Computing and Network Security (ICSNS), Coimbatore, India, 14–16 February 2018; pp. 1–6. [CrossRef]

- 55. Rastgoo, S.; Mahdavi, Z.; Azimi Nasab, M.; Zand, M.; Padmanaban, S. Using an Intelligent Control Method for Electric Vehicle Charging in Microgrids. *World Electr. Veh. J.* 2022, 13, 222. [CrossRef]
- 56. Poursafar, N.; Taghizadeh, S.; Hossain, M.J.; Blaabjerg, F. An Enhanced Control Strategy for an Ultra-Fast EV Charging Station in a DC Microgrid. *Int. J. Electr. Power Energy Syst.* 2023, 146, 108727. [CrossRef]
- Zhaoxia, X.; Haodong, F.; Guerrero, J.M.; Hongwei, F. Hierarchical Control of a Photovoltaic/Battery Based DC Microgrid including Electric Vehicle Wireless Charging Station. In Proceedings of the IECON 2017—43rd Annual Conference of the IEEE Industrial Electronics Society, Beijing, China, 29 October–1 November 2017; pp. 2522–2527. [CrossRef]
- Bhargavi, K.M.; Jayalakshmi, N.S. A New Control Strategy for Plug-in Electric Vehicle of DC Microgrid with PV and Wind Power Integration. J. Electr. Eng. Technol. 2019, 14, 13–25. [CrossRef]
- Senapati, M.K.; Al Jaafaari, K.; Al Hosani, K.; Muduli, U.R. Flexible Control Approach for DC Microgrid Oriented Electric Vehicle Charging Station. In Proceedings of the 2023 IEEE IAS Global Conference on Renewable Energy and Hydrogen Technologies (GlobConHT), Male, Maldives, 11–12 March 2023; pp. 1–6. [CrossRef]
- 60. Jiménez Carrizosa, M.; Iovine, A.; Damm, G.; Alou, P. Droop-Inspired Nonlinear Control of a DC Microgrid for Integration of Electrical Mobility Providing Ancillary Services to the AC Main Grid. *IEEE Trans. Smart Grid* **2022**, *13*, 4113–4122. [CrossRef]
- Qin, D.; Sun, Q.; Wang, R.; Ma, D.; Liu, M. Adaptive Bidirectional Droop Control for Electric Vehicles Parking with Vehi-cle-to-Grid Service in Microgrid. CSEE J. Power Energy Syst. 2020, 6, 793–805. [CrossRef]
- 62. Rajamand, S. Feedback-Based Control Structure for Frequency/Voltage Regulation Using the State of Electrical Vehicle Charge Station and Point Estimation Method. *Sustain. Energy Technol. Assess.* **2022**, *51*, 101922. [CrossRef]
- Benamar, A.; Travaillé, P.; Clairand, J.-M.; Escrivá-Escrivá, G. Non-Linear Control of a DC Microgrid for Electric Vehicle Charging Stations. Int. J. Adv. Sci. Eng. Inf. Technol. 2020, 10, 593–598. [CrossRef]
- Marinescu, A.; Serban, I. A Smart Residential Microgrid Based on Renewable Energy Sources with Integrated Electric Vehicle Charging Station. In Proceedings of the 2018 International Symposium on Fundamentals of Electrical Engineering (ISFEE), Bucharest, Romania, 1–3 November 2018; pp. 1–5. [CrossRef]
- 65. Hao, Y.; Dong, L.; Liang, J.; Liao, X.; Wang, L.; Shi, L. Power Forecasting-Based Coordination Dispatch of PV Power Generation and Electric Vehicles Charging in Microgrid. *Renew. Energy* **2020**, *155*, 1191–1210. [CrossRef]
- Da Costa Mendes, P.R.; Normey-Rico, J.E.; Alba, C.B. Economic Energy Management of a Microgrid Including Electric Vehicles. In Proceedings of the 2015 IEEE PES Innovative Smart Grid Technologies Latin America (ISGT LATAM), Montevideo, Uruguay, 5–7 October 2015; pp. 869–874. [CrossRef]
- Mendes, P.R.C.; Maestre, J.M.; Bordons, C.; Normey-Rico, J.E. Binary Search Algorithm for Mixed Integer Optimization: Application to Energy Management in a Microgrid. In Proceedings of the 2016 European Control Conference (ECC), Aalborg, Denmark, 29 June–1 July 2016; pp. 2620–2625. [CrossRef]
- 68. Liu, Y.; Wang, Y.; Li, Y.; Gooi, H.B.; Xin, H. Multi-Agent Based Optimal Scheduling and Trading for Multi-Microgrids Integrated With Urban Transportation Networks. *IEEE Trans. Power Syst.* **2021**, *36*, 2197–2210. [CrossRef]
- 69. Jiao, F.; Zou, Y.; Zhang, X.; Zhang, B. Online optimal dispatch based on combined robust and stochastic model predictive control for a microgrid including EV charging station. *Energy* **2022**, 247, 123220. [CrossRef]
- Alkhafaji, M.; Luk, P.; Economou, J. Optimal Design and Planning of Electric Vehicles Within Microgrid. In Proceedings of the Advanced Computational Methods in Energy, Power, Electric Vehicles, and Their Integration (ICSEE LSMS 2017), Singapore, 22–24 September 2017; Communications in Computer and Information Science. Springer: Singapore; Volume 763. [CrossRef]
- 71. Han, X.; Li, X.; Wang, Z. An Optimal Control Method of Microgrid System with Household Load Considering Battery Service Life. *J. Energy Storage* **2022**, *56*, 106002. [CrossRef]
- Chen, C.; Duan, S. Microgrid Economic Operation Considering Plug-in Hybrid Electric Vehicles Integration. J. Mod. Power Syst. Clean. Energy 2015, 3, 221–231. [CrossRef]
- Yuan, H.; Wei, G.; Zhu, L.; Zhang, X.; Zhang, H.; Luo, Z.; Hu, J. Optimal Scheduling for Micro-grid Considering EV Charging-Swapping-Storage Integrated Station. *IET Gener. Transm. Distrib.* 2020, 14, 1127–1137. [CrossRef]
- Xu, T.; Sun, H.; Zhu, B.; Long, Y.; Wang, H.; Li, Z. Economic Optimization Control of Microgrid with Electric Vehicles. In Proceedings of the 2018 5th International Conference on Information Science and Control Engineering (ICISCE), Zhengzhou, China, 20–22 July 2018; pp. 733–736. [CrossRef]
- Cheikh-Mohamad, S.; Sechilariu, M.; Locment, F. Real-Time Power Management Including an Optimization Problem for PV-Powered Electric Vehicle Charging Stations. *Appl. Sci.* 2022, 12, 4323. [CrossRef]
- 76. Edathil, S.; Singh, S.P. Energy Regulation of EV Using MILP for Optimal Operation of Incentive Based Prosumer Microgrid with Uncertainty Modelling. *Int. J. Electr. Power Energy Syst.* **2022**, *134*, 107353. [CrossRef]
- Bracco, S.; Delfino, F.; Pampararo, F.; Robba, M.; Rossi, M. A Dynamic Optimization-Based Architecture for Polygeneration Microgrids with Tri-Generation, Renewables, Storage Systems and Electrical Vehicles. *Energy Convers. Manag.* 2015, *96*, 511–520. [CrossRef]
- Yan, J.; Menghwar, M.; Asghar, E.; Panjwani, M.K.; Liu, Y. Real-Time Energy Management for a Smart-Community Microgrid with Battery Swapping and Renewables. *Appl. Energy* 2019, 238, 180–194. [CrossRef]
- Liao, J.T.; Lin, C.I.; Chien, C.Y.; Yang, H.T. The Distributed Energy Resources Operation for EV Charging Stations and SHEMS in Microgrids. In Proceedings of the 2014 International Conference on Intelligent Green Building and Smart Grid (IGBSG), Taipei, Taiwan, 23–25 April 2014; pp. 1–6. [CrossRef]

- Wang, T.; O'Neill, D.; Kamath, H. Dynamic Control and Optimization of Distributed Energy Resources in a Microgrid. *IEEE Trans. Smart Grid* 2015, 6, 2884–2894. [CrossRef]
- Mendes, P.R.C.; Normey-Rico, J.E.; Bordons, C. Distributed Energy Management System for V2G Networked Microgrids. In Proceedings of the 2017 IEEE PES Innovative Smart Grid Technologies Conference—Latin America (ISGT Latin America), Quito, Ecuador, 20–22 September 2017; pp. 1–6. [CrossRef]
- Mendes, P.R.C.; Valverde Isorna, L.; Bordons, C.; Normey-Rico, J.E. Energy management of an experimental microgrid coupled to a V2G system. J. Power Sources 2016, 327, 702–713. [CrossRef]
- Asaad, A.; Ali, A.; Mahmoud, K.; Shaaban, M.F.; Lehtonen, M.; Kassem, A.M.; Ebeed, M. Multi-objective optimal planning of EV charging stations and renewable energy resources for smart microgrids. *Energy Sci. Eng.* 2023, 11, 1202–1218. [CrossRef]
- 84. Iqbal, S.; Xin, A.; Jan, M.U.; Salman, S.; Zaki, A.u.M.; Rehman, H.U.; Shinwari, M.F.; Abdelbaky, M.A. V2G Strategy for Primary Frequency Control of an Industrial Microgrid Considering the Charging Station Operator. *Electronics* **2020**, *9*, 549. [CrossRef]
- Datta, U.; Kalam, A.; Shi, J.; Li, J. Electric Vehicle Charging Station for Providing Primary Frequency Control in Microgrid. In Proceedings of the 2019 14th IEEE Conference on Industrial Electronics and Applications (ICIEA), Xi'an, China, 19–21 June 2019; pp. 2440–2444. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.